Author's response to reviews

Title: Glucagonoma syndrome: a case report

Authors:

Pablo Granero Castro (pgranerocastro@aecirujanos.es)
Alberto Miyar de León (albertmiyar@yahoo.es)
Jose Granero Trancón (jose.granero@aecirujanos.es)
Paloma Álvarez Martínez (dra.palialvarez@gmail.com)
Jose A Álvarez Pérez (josealvar@telecable.es)
Jose C Fernández Fernández (blendio@terra.es)
Carmen M García Bernardo (cmgberardo2@hotmail.com)
Luis Barneo Serra (lbarneo@uniovi.es)
Juan J González González (jjglezglez@gmail.com)

Version: 2 Date: 29 March 2011

Author's response to reviews: see over
Glucagonoma syndrome: a case report

Manuscript ID: 1273909939521873

Dear Editor:

We analyzed with attention the reviewer’s observations and we think that most of them are very appropriate to improve the work.

I am sending you the manuscript with all the modifications made according to the reviewer’s comments.

I am sending you the following items:

1. Marked-up version with all track changes visible (underlined text).
2. Point-by-point reply to all queries raised by the reviewers.

CHANGES AND CORRECTIONS MADE AFTER REVIEWER’S COMMENTS

- **Introduction:** It has been rewritten. Paragraph 1 and 3 of the earlier discussion and their references have been included in the Paragraph 1 of the new Introduction.

- **Case presentation:** The ethnicity of the patient has been added in the Case presentation section.

- **Discussion:**
  - Paragraph 2 of the Discussion (historical approach to the glucagonoma syndrome) has been deleted.
  - A sentence concerning the differential diagnosis of glucagonoma has been added in paragraph 5 of the Discussion along with its corresponding references.
  - A sentence concerning the rare pancreatic tumours has been included in the paragraph 5 of the Discussion along with its corresponding references.

- No acknowledgement section is required in this manuscript.

Yours faithfully.

Pablo Granero Castro